Author's response to reviews

Title: In-vivo visualisation of the anatomical structures related to the acupuncture points Dai mai and Shen mai by MRI: A single-case pilot study

Authors:
Roy Moncayo (anmeldung@womed.at)
Ansgar Rudisch (a.rudisch@uibk.ac.at)
Christian Kremser (christian.kremser@uibk.ac.at)
Markus Diemling (markus@nud.se)

Version: 7 Date: 9 March 2007

Author's response to reviews:
Changes have been made according to the instructions of the production unit.